Author's response to reviews

Title: A model to estimate the cost of the National Essential Public Health Services Package in Beijing, China

Authors:

Delu Yin (deluyin@126.com)
Sabrina T Wong (sabrina.wong@nursing.ubc.ca)
Wei Chen (cwei214@vip.sina.com)
Qianqian Xin (xingqianqian1014@126.com)
Lihong Wang (wlh3000@163.com)
Mingming Cui (bicmm@126.com)
Tao Yin (gouweicao2911@163.com)
Ruili Li (chacreally@163.com)
Xiaoguo Zheng (Xiaoguo_zheng@163.com)
Huiming Yang (yanghuimin85@126.com)
Juanjuan Yu (wjujuw@aliyun.com)
Weizhong Yang (yangwz@chinacdc.cn)
Bowen Chen (1726137684@qq.com)

Version: 2
Date: 22 November 2014

Author's response to reviews:

line number, list of abbreviations were added, and email address were changed.